IL-33/ST2 pathway regulates neutrophil migration and predicts outcome in patients with severe alcoholic hepatitis

Graphical abstract



Highlights

- The IL-33/ST2 pathway is defective in the neutrophils of patients with severe alcoholic hepatitis.
- This defect is associated with a higher risk of developing infection.
- Dosage of the decoy receptor sST2 helps identify patients at risk of death and/or infection.
- The defect in IL-33/ST2 in neutrophils is responsible for lower migration capacities.
- Treating neutrophils with recombinant IL-33 restores, at least in part, their migration capacities.

Authors

Florent Artru, Mohamed Bou Saleh, François Maggiotto, ..., Philippe Mathurin, Laurent Dubuquoy, Alexandre Louvet

Correspondence

alexandre.louvet@chru-lille.fr (A. Louvet), laurent. dubuquoy@inserm.fr (L. Dubuquoy).

Lay summary

The neutrophils of patients with severe alcoholic hepatitis are associated with a defect in the IL-33/ ST2 pathway. This defect is associated with lower migration capacities in neutrophils and a higher probability of getting infected. Administration of IL-33 to the neutrophils at least partly restores this defect and may be effective at reducing the risk of infection in patients with severe alcoholic hepatitis.



IL-33/ST2 pathway regulates neutrophil migration and predicts outcome in patients with severe alcoholic hepatitis^{*}

 Florent Artru^{1,2}, Mohamed Bou Saleh², François Maggiotto², Guillaume Lassailly^{1,2}, Massih Ningarhari^{1,2}, Julie Demaret^{2,3}, Line-Carolle Ntandja-Wandji^{1,2},
Jean-Paul Pais de Barros⁴, Julien Labreuche⁵, Elodie Drumez⁵, Doumet Georges Helou^{6,7}, Sébastien Dharancy^{1,2}, Emilie Gantier², Axel Périanin^{8,9,10}, Sylvie Chollet-Martin^{6,7}, Ramon Bataller¹¹, Philippe Mathurin^{1,2}, Laurent Dubuquoy^{2,*}, Alexandre Louvet^{1,2,*}

¹Hôpital Claude-Huriez, Service Maladies de l'Appareil Digestif, CHU Lille, Lille, France; ²Université de Lille/Inserm/CHU de Lille, U995 - LIRIC – Lille Inflammation Research Center, Lille, France; ³Centre de Biologie-Pathologie, CHU de Lille, Lille, France; ⁴INSERM, LNC UMR 1231, Université de Bourgogne Franche-Comté, Dijon, France; ⁵Département de biostatistiques, CHU de Lille, Lille, France; ⁶Inserm/Université Paris-Sud/Université Paris-Saclay, UMR996, Chatenay-Malabry, France; ⁷Assistance publique-Hôpitaux de Paris, Hôpital Bichat, Laboratoire d'immunologie « Autoimmunité et Hypersensibilités », Paris, France; ⁸Inserm/Faculté de Médecine Xavier Bichat, UMRS-1149, Paris, France; ⁹CNRS, ERL-8252 Centre de Recherche sur l'Inflammation, Paris, France; ¹⁰Université Paris Diderot, Sorbonne Paris Cité, Laboratoire d'Excellence INFLAMEX, Paris, France; ¹¹Division of Gastroenterology, Hepatology and Nutrition, Pittsburgh Liver Research Center, University of Pittsburgh, PA, USA

Background & Aims: Severe alcoholic hepatitis (SAH) is associated with a high risk of infection. The IL-33/ST2 pathway is involved in sepsis control but data regarding its role in alcohol-related liver disease (ALD) are lacking. We aimed to characterize the role of IL-33/ST2 in the polymorphonuclear neutrophils (PMNs) of patients with ALD and SAH.

Methods: Serum and circulating neutrophils were collected from patients with SAH, alcoholic cirrhosis and healthy controls. We quantified IL-33/ST2 pathway activity and CXCR2 at baseline and after exposure to IL-33. We also determined the migration capacity of PMNs.

Results: The decoy receptor of IL-33 (soluble ST2 [sST2]) was increased in SAH *vs.* cirrhosis and controls, demonstrating the defect in this pathway during ALD. The sST2 level was associated with response to treatment, 2-month survival, infection-free survival and probability of infection in SAH. Endotoxemia was weakly correlated with sST2. GRK2, a negative regulator of CXCR2, was overexpressed in PMNs of patients with SAH and cirrhosis and was decreased by IL-33. CXCR2 levels on PMNs were lower in SAH *vs.* cirrhosis and controls. Treatment with IL-33 partially restored CXCR2 expression in SAH and cirrhosis. PMN migration upon IL-8 was lower in patients with SAH and cirrhosis *vs.* controls. Treatment with IL-33 partially restored migration in those with SAH and cirrhosis. Interestingly, the migration capacity of PMNs and the response to IL-33 were

* Guest Editor: Patrick Kamath.

https://doi.org/10.1016/j.jhep.2019.12.017



Conclusion: The IL33/ST2 pathway is defective in SAH and predicts outcome. This defect is associated with decreased CXCR2 expression on the surface of PMNs and lower migration capacity, which can be corrected by IL-33, especially in patients responding to steroids. These results suggest that IL-33 has therapeutic potential for SAH and its infectious complications. **Lay summary:** The neutrophils of patients with severe alcoholic hepatitis are associated with a defect in the IL-33/ST2 pathway. This defect is associated with lower migration capacities in neutrophils and a higher probability of getting infected. Administration of IL-33 to the neutrophils at least partly restores this defect and may be effective at reducing the risk of infection in patients with severe alcoholic hepatitis.

© 2020 Published by Elsevier B.V. on behalf of European Association for the Study of the Liver.

Introduction

Severe alcoholic hepatitis (SAH) is a life-threatening condition associated with short-term liver failure and a high risk of death. SAH is mainly defined by a Maddrey's discriminant function $\geq 32^{1}$ and cirrhosis is observed in more than 90% of cases.^{2,3} The only treatment that has been shown to be effective in reducing shortterm mortality (*i.e.* at 28 days) is prednisolone, 40 mg/day for 1 month.^{4–6} In case of severe alcoholic hepatitis, the estimated short- and medium-term probability of death is around 20–30%. The Lille model is a helpful prognostic score which identifies patients that do not respond to medical treatment and who are at high risk of death.⁷ This model ranges from 0 to 1 and patients with a score >0.45 are classified as non-responders. There is no alternative pharmaceutical treatment for these patients to date and their outcome is poor.

Bacterial infections are one of the main drivers of the increased risk of mortality in severe AH. An estimated 25% of patients with SAH are admitted with infection at diagnosis and





Keywords: Alcoholic hepatitis; Cirrhosis; Infection; Polymorphonuclear neutrophils; Interleukin-33; Migration.

Received 15 July 2019; received in revised form 30 November 2019; accepted 12 December 2019; available online 15 January 2020

^{*} Corresponding authors. Address: Service des maladies de l'appareil digestif and Unité INSERM 995, Hôpital Huriez, Rue Polonovski, F- 59037 Lille cedex. Tel.: 33 3 20 44 53 03, fax: 33 3 20 44 55 64.

E-mail addresses: alexandre.louvet@chru-lille.fr (A. Louvet), laurent.dubuquoy @inserm.fr (L. Dubuquoy).

another 25% will develop infection during treatment with corticosteroids.⁸ Thus, targeting infection in patients with SAH is an attractive area of drug development.

There are multiple mechanisms associated with the increased risk of infection during alcohol-related liver disease (ALD), which involve different levels of the innate and adaptive immune system. For example, the antigen presentation function of antigen presenting cells is downregulated, monocyte and macrophage bacterial killing is reduced and T lymphocyte interferon (IFN) production in response to lipopolysaccharide (LPS) is reduced.⁹ The presence of a profound immune dysfunction associated with a high level of circulating neutrophils in AH may seem paradoxical. Some studies have reported an increased resting burst in circulating polymorphonuclear neutrophils (PMNs) which is in contrast with a reduced oxidative burst induced by E. coli and a decrease in phagocytosis capacities which has been suggested by some authors.¹⁰⁻¹² Another key function of PMNs is the migration to the site of infection, which seems to be impaired in decompensated alcohol-related cirrhosis.¹³ When considering the crucial role of PMNs in the resolution of sepsis, studies to understand which pathways are involved in impairment of PMN function, especially migration, are essential. The IL-33/ST2 pathway is a potential target that could play a role in the high rate of infections in alcoholic liver disease, in particular in severe AH.

IL-33 is a member of the IL-1 family that is mainly located in the nucleus of barrier cells, such as epithelial cells and endothelial cells, but also in several organs, such as the lungs, skin, spleen and liver.^{14–17} IL-33 participates in the regulation of gene expression but is also considered a damage-associated molecular pattern (DAMP) released after cell injury and/or necrosis especially in tissue macrophages.¹⁸ Binding of IL-33 to the transmembrane form of ST2 (suppression of tumorigenicity 2) (ST2L) leads to subsequent activation of multiple intracellular signaling pathways which depend on cell type and location. In addition to IL-33/ST2 signaling on the cell surface, the ST2 gene also encodes a soluble form of the protein (sST2) lacking the transmembrane domain and acting as a decoy receptor for IL-33, inhibiting signal transduction of this pathway. In liver diseases, the IL-33/ST2 pathway has been involved in the pathogenesis of liver injury related to viral hepatitis, fatty liver disease, acute hepatitis and has been suggested to play a role in fibrogenesis.¹⁹ It has been shown that the IL-33/ST2 pathway attenuates sepsis in the specific setting of infection without liver impairment by enhancing neutrophil influx to the site of infection (i.e. migration) via increased expression of chemokine receptor CXCR2 on circulating PMNs in animal models and *ex vivo* experiments.²⁰ Based on the relationship between IL-33/ST2 pathway activation and neutrophil migration we hypothesize that this pathway could play a role in the PMN immune dysfunction observed in alcoholic liver disease and severe AH.

The aims of our study were i) to characterize the role of the IL-33/ST2 pathway in the PMN of patients with ALD and severe AH; ii) to evaluate the impact of IL-33/ST2 pathway modulation on circulating PMN function *ex vivo*, especially migration capacities.

Patients and methods

Patients

This study included patients with severe alcoholic hepatitis (SAH), decompensated cirrhosis and healthy controls. Individuals

with SAH were included from a prospective cohort of consecutive patients (n = 161), managed in the hepatology department of Hôpital Claude-Huriez, Lille, France, Alcoholic hepatitis was clinically suspected using the classical diagnostic criteria. *i.e.* recent onset of jaundice (less than 3 months), heavy alcohol consumption, clinical signs of hepatic decompensation (ascites, encephalopathy), elevated aminotransferases (less than 300 IU/L with aspartate aminotransferase >alanine aminotransferase) and high serum bilirubin levels (greater than 3.0 mg/dl) with no other identifiable cause of liver disease.²¹ Alcoholic hepatitis was confirmed by liver biopsy in 148 patients (91.9% of cases) and diagnosis was based on the presence of the following criteria: hepatocellular damage (presence of Mallory bodies and hepatocellular ballooning), PMN infiltrate and steatosis associated with fibrosis.²¹ In the remaining cases, liver histology was not available in 13 patients for the following reasons: 4 patients due to severe alterations in hemostasis contraindicating liver biopsy, 2 patients due to the presence of a transjugular intrahepatic portosystemic shunt and 7 patients for a procedure failure. Liver biopsy was performed according to standard clinical practice in France and not for research purposes. SAH was defined as a Maddrey's discriminant function \geq 32 and the severity of liver injury was also assessed by the model for end-stage liver disease (MELD) and Child-Pugh scores. After liver biopsy, 154 (95.6%) patients were treated with prednisolone (40 mg/day) for 7 days and therapeutic response was assessed by the Lille model.⁷ Seven patients were not treated with prednisolone because of very severe liver dysfunction which is associated with a very low probability of response to steroids. A non-response to medical treatment was defined by a Lille model >0.45 according to the ideal cut-off published in the original paper.⁷ The response to medical management in non-treated patients was also assessed by the Lille model as this model has also been shown to be effective in placebo-treated patients.³⁻⁵ Systematic screening of infection was performed at admission and consisted of chest X-ray and blood, urine, and ascites cultures as published by our group.⁸ During the follow-up, the same infection screening as that of admission was renewed in case of clinical or biologic signs of infection. Serum and plasma samples were collected before the initiation of treatment and at day 7 in patients treated with prednisolone. Serum and plasma samples were collected at diagnosis of SAH and 7 days thereafter in non-treated patients. It should be noted that patients were not included in the study if they developed sepsis between 2 blood samples.

Clinical outcome

Patients were followed-up for at least 2 months. Survival in patients with SAH was calculated from the first day of treatment with prednisolone to the last follow-up or liver transplantation (n = 4). Survival in patients with decompensated cirrhosis was calculated from the time of blood sample collection. The status (alive or dead) of lost to follow-up patients was assessed by telephoning a family member, general practitioner, or both, or by contacting the death registry at the patient's birthplace. We prospectively recorded all infectious events to evaluate the incidence of infection in patients with SAH.

Analyses performed

Two types of analyses were performed in patients with SAH: cellular analyses on fresh neutrophils obtained from 2012 to 2019 (see below) and serum analyses on frozen samples

collected from 2006 to 2012 before the initiation of the study and then prospectively from 2012 to 2019 to increase the sample size. Diagnostic criteria and patient management were the same in the 2 cohorts of patients.

Patients with SAH were compared to patients with decompensated alcoholic cirrhosis but without superimposed SAH admitted to our unit during the study period. The diagnosis of cirrhosis was based on either liver biopsy, clinical, laboratory and/or imaging features. We also included healthy controls during the same study period.

The procedure for blood sample collection (cells, serum, plasma) was the same in patients with SAH and cirrhosis, and healthy controls. To collect plasma and serum, blood samples were centrifuged at 1,500 rpm for 10 min at 4°C. We used the MACSxpress Neutrophil isolation human kit (ref. 130104434, Miltenyi Biotec, Germany) to isolate PMNs. Briefly, 4.5 ml of whole blood were incubated with 2 ml of manufacturer's buffer in Falcon 15 ml tubes for 5 min on the Macsmix rotator at 12 rpm. The tube was then placed on the MACSxpress Separator for 15 min at room temperature. The plasma supernatant was then collected with high purity PMNs. PMNs were incubated in the presence or absence of 50 ng/ml of recombinant human IL-33 50 ng/ml (ref. 3625-IL, Bio-Techne USA) for 30 min at 37°C. Blood samples were collected when prednisolone was initiated (day 0 -D0) and 7 days thereafter (day 7 - D7) in the SAH group and at admission to the cirrhosis group.

All patients agreed to participate in the study and written consent was obtained before analyses were performed. The study was registered under the number CPP14/67 at the "Comité de Protection des Personnes" of the Hauts de France région.

IL-33 and sST2 level

We assessed IL-33 and sST2 serum concentrations by ELISA using human IL-33 ELISA kit (ref. ADI-900-201, ENZO Life Sciences, USA) and ST2 human kit (ref. DY523B-05, Bio-Techne USA). Duplicate measurements were performed for each sample.

Quantification of endotoxemia

Endotoxemia was assessed using HPLC-MS/MS (high-performance liquid chromatography coupled with mass spectrometry) to detect 3-hydroxymyristate (3-HM), a lipid component of LPS, in the serum of controls, patients with cirrhosis and patients with SAH. Methods are described in detail in Weil *et al.* and Pais de Barros *et al.*^{22,23}

Immunocytochemistry for GRK2

Two-hundred thousand PMNs were fixed with cold methanol for 30 s on Cytospin slides. Slides were incubated (or not as negative controls) overnight at 4°C with primary anti-GRK2 polyclonal rabbit antibody (ref. 137666, Abcam, U.K) and stained with secondary FITC-conjugated goat antibody to rabbit (ref. A11034 Life technologies, USA). We captured images with a DM5500B microscope (Leica, Germany) equipped with DFC310 FX camera (Leica, Germany) and analyzed them by Leica Application Suite (LAS, Leica, Germany). All the cells from at least 5 randomly chosen fields in each slide performed in duplicate, were analyzed from at least 2 individual experiments. GRK2 expression was quantified in parallel for the entire set after defining thresholds using ImageJ software.²⁴ Correction for uneven illumination in fluorescence images was performed at the same time. Results are expressed as mean intensity with relative units (MIRU).

Flow cytometry analysis

CXCR2 expression

After isolation, we pretreated PMNs with IL-33 (50 ng/ml) or HBSS as described above for 30 min at 37°C. Surface staining of circulating PMNs with antibodies to CXCR2 (12-1829-42, eBiosciences) for 30 min at 4°C was performed and cells were then analyzed by a Cyan ADP (Beckman Coulter, USA). For each condition, an isotype control tube (with mouse IgG1 k ISO control, eBiosciences) was used for non-specific signal subtraction.

SIGIRR expression

Surface staining of circulating PMNs was performed using APClabeled anti-SIGIRR (FAB990A, R&D Systems) for 15 min at room temperature. Cytometry analyses were performed on a Navios flow cytometer using the Navios software (Beckman Coulter).

Apoptosis assessment

After isolation, we performed pretreatment of PMNs by IL-33 (50 ng/ml) or HBSS as described above for 30 min at 37°C. According to manufacturer's instructions, cells were washed with PBS and suspended in Annexin V Binding Buffer (FITC Annexin V Apoptosis Detection Kit with 7 AAD, 640922, Biolegend). Cell staining was performed using FITC labeled Annexin V and 7-AAD viability staining solution. After 15 min of incubation at room temperature, Annexin V Binding Buffer was added prior to cell analysis on a Navios flow cytometer. Intact cells were lacking both expression of Annexin V and 7-AAD and apoptotic cells were considered as cells expressing Annexin V but with low expression of 7-AAD.

Neutrophil functional tests

Migration. The migration capacity of PMN was assessed *ex vivo*.^{25–27} Briefly, after isolation of PMNs 1×10^6 cells were deposited in the insert (polycarbonate filter 3 µm) of transwell kits (ref. 3415, Corning, USA). Five nM of recombinant human IL-8 (ref. 208-IL, Bio-Techne USA) or 5 nM of recombinant human IL-8 plus IL-33 50 ng/ml in HBSS (red phenol, calcium and magnesium free) were added to the lower well. PMNs were then incubated for 1 h at 37° C, 5% CO₂. The insert was removed, the culture medium containing cells was collected and the number of cells were counted by FACS (LSR FORTESSA X20 cytometer (BD Biosciences, USA).

Phagocytosis. The Phagotest (ref. 341060, BD Bioscience) was used to measure phagocytosis using FITC-labeled opsonized *E. coli* bacteria. The mean fluorescence intensity of the respective antibodies on neutrophils was analyzed by FACS analysis with LSR FORTESSA X20 cytometer (BD Biosciences, USA).

Statistical analysis

Quantitative variables were expressed as medians with interquartile ranges. Categorical variables were expressed in numbers and percentages. Comparisons between quantitative variables were performed by the Mann-Whitney *U* test, paired-sample Student's *t* test or Kruskal-Wallis, as appropriate. Correlations between quantitative variables were performed by calculating the Spearman's rank correlation coefficient. Strength of correlation was defined using the Spearman' coefficient (r): strong correlation if r >0.75, moderate correlation if r >0.5 and weak if r <0.5.²⁸ Two-month overall survival was calculated by the

Table 1. Baseline characteristics of patients with severe	alcoholic hepatitis and cirrhosis in	cluded in prognostic analyses.
---	--------------------------------------	--------------------------------

	SAH	Cirrhosis	p value
Number	119	24	
Male/Female, n (%)	82/37 (68.9/31.1)	19/5 (79.2/27.8)	0.3
Age, years	50.4 (41.8-58.1)	60.5 (52.7-66.9)	0.0001
Ascites, n (%)	71 (59.7)	21 (87.5)	0.05
Hepatic encephalopathy, n (%)	19 (16.2)	10 (41.6)	0.01
Leukocytes, G/L	11.2 (8.6–14.5)	9.1 (6.1–11.7)	0.09
CRP, mg/L	28.5 (20.5-43)	13.5 (6–17.3)	0.03
Albumin, g/L	27 (24–31)	33.5 (30.5–38.7)	0.002
Bilirubin, mg/dl	17 (11.3–26.9)	4.4 (1.7–6.6)	<0.0001
Creatinine, mg/dl	0.8 (0.6–1.1)	1.4 (1–1.5)	0.001
Prothrombin rate, %	40 (33–51)	47 (34–60)	0.06
MELD score	25 (22-31)	19 (16–23)	0.001
Child-Pugh score	11 (9–12)	9 (7–11)	0.01
Lille model	0.25 (0.08-0.58)	n.a.	
Responders according to 0.45, N (%)	75 (63)	n.a.	
2-month survival, %	75.5 (67.3-83.7)	65.3 (45.8-84.8)	0.12

3-HM, 3-hydroxymyristate; MELD, model for end-stage liver disease; SAH, severe alcoholic hepatitis; sST2, soluble ST2.

The characteristics detailed below are those of patients included in the serum analyses (IL-33, sST2 and 3-HM) performed to determine the relationship between the IL33-ST2 pathway with prognosis and infection. Data are presented as median (IQR). Comparisons between quantitative variables were performed by the Mann-Whitney *U* test or Kruskal-Wallis, as appropriate.

Kaplan-Meier method and intergroup comparison according to sST2 tertiles was done using the log-rank test. We assessed the association between sST2 (treated as a continuous variable) and endpoints (2-month mortality or a composite of mortality and infection within the first 2 months) using Cox's proportional hazard regression model. The association between sST2 and infection at 2 months was assessed using Fine and Gray's model, treating death as a competing risk. We derived effect sizes from Cox's and Fine and Gray's models by calculating hazard ratio (HR) per 1 SD increase in sST2, as well as discrimination values by calculating the Harrell's c-index of agreement (using the approach of Wolbers and colleagues in the presence of competing risks for infection outcome²⁹). Finally, we drew ROC curves for 2-month prediction of overall and infection-free survival analyses. The significance level was set at 0.05 with a 2sided test. All statistical analyses were performed using NCSS 2011 or SAS (version 9.4, SAS Institute, Cary, NC) software.

Results

A total of 161 patients with SAH, 72 with severe cirrhosis and 28 healthy controls were included in the different analyses in the study. The main patient characteristics at inclusion are reported in Table S1. As expected, compared to patients with cirrhosis, patients with SAH were younger (52.4 [42.6–58.6] vs. 58.2 [51.5–63.7] years, p < 0.0001), and more frequently had parameters indicative of systemic inflammatory response syndrome (leukocyte count at 11.7 [8.7–15.1] vs. 8.6 [5.1–11.1] G/L, p = 0.0005 and CRP at 29 [21–39] vs. 12 [9–17] mg/L, p = 0.0006). Bilirubin and MELD score were higher in the SAH group compared to the cirrhosis group (17.4 [10.9–29.1] vs. 5 [3.2–7.7] mg/dl, p < 0.0001; and 25 [22–31] vs. 20 [17–26], p < 0.0001, respectively); however, Child-Pugh score (11 [10-12] vs. 10 [8-13], p = 0.3) and 2-month survival (71.9 [65–78.8] vs. 66.6 [55.5–77.8], p = 0.22) were not different between the 2 groups.

IL-33/ST2 pathway is impaired in patients with SAH

A total of 119 patients with SAH, 24 with severe cirrhosis and 12 healthy controls were included in the serum analysis part of the study. The main characteristics at inclusion are reported in

Table 1 and are very close to the characteristics of patients withSAH described above.

Circulating IL-33 levels were relatively low in the 3 groups and we observed no significant difference between IL-33 levels of controls (91.7 pg/ml; IQR 80.5-96.5), or patients with cirrhosis (86 pg/ml; IQR 77.1-90.8) or SAH (107.7 pg/ml; IQR 80-147), p = 0.12 for comparison between control and SAH and p = 0.11 between cirrhosis and SAH (Fig. 1A). In contrast, there were marked differences in blood levels of the IL-33 decoy receptor (sST2), known to be the regulator of this pathway, between the 3 groups. The level of sST2 was increased in SAH (415,062 pg/ml, 231,461-691,729) compared to cirrhosis (191,511 pg/ml, 126,316-316,040, p <0.0001) and control (16,545 pg/ml, 11,645–22,162, p <0.0001). These results strongly suggest an attenuated IL-33/ST2 pathway in SAH compared to the other groups (Fig. 1B). We then assessed the relationship between IL-33 and sST2 level and the response to steroid treatment according to the 0.45 cut-off of the Lille model. At baseline (initiation of prednisolone), serum levels of IL-33 were not different between responders (*i.e.* Lille <0.45) and non-responders (*i.e.* Lille ≥0.45): 104.1 pg/ml (81.2–127.8) vs. 119.2 pg/ml (78.2-164.2), p = 0.3 (Fig. 1C). In contrast, responders had lower sST2 levels at baseline than nonresponders: 336,775 pg/ml (205,896-636,541) vs. 598,788 pg/ ml (270,260–708,915), p = 0.01 (Fig. 1D). These important results indicate that serum sST2 levels could be useful to predict response to prednisolone.

In SAH group, IL-33 levels were not different between D0 and D7 either on overall patients (107.7 vs. 109.2 pg/ml, p = 0.2), or in responders (104.1 vs. 106.6 pg/ml, p = 0.5) or non-responders (119.2 vs. 109.3 pg/ml, p = 0.2). sST2 levels increased between D0 and D7 in the SAH group from 415,062 to 656,775 pg/ml, p < 0.0001. Such increases were observed in both groups of responders and non-responders (336,775 vs. 589,500, p < 0.0001 and 598,788 vs. 761,656, p < 0.0001 pg/ml, respectively).

sST2 predicts survival in patients with SAH and cirrhosis

The overall 2-month survival in patients with SAH and cirrhosis, included in serum analysis of the IL-33/ST2 pathway, was 76.1%



Fig. 1. IL-33/ST2 pathway expression and association with survival. (A, B) Serum levels of IL-33 (A) and sST2 (B) from healthy controls (n = 12), patients with cirrhosis (n = 24) and patients with SAH (n = 119). Results are expressed as median in pg/ml with IQR. (C, D) IL-33/ST2 pathway expression according to response to medical treatment assessed by the Lille score with responders (n = 75) defined by a score <0.45 and non-responders (n = 44) by a score \ge 0.45. (E) 2-month survival according to percentile distribution ($\leq 33^{\text{th}}$ [n = 39], between 33^{th} and 66^{th} [n = 40], $\geq 66^{\text{th}}$ [n = 40]) level of sST2 at D0 in patients with SAH. (F) ROC curve in binormal representation of the capacity of sST2 to predict 2month survival in patients with SAH (n = 119). Comparisons between quantitative variables were performed by the Mann-Whitney U test or Kurskall-Wallis, as appropriate. Two-month overall survival was calculated by the Kaplan-Meier method and intergroup comparison according to sST2 tertiles was done by the log-rank test. n.s., non significant, *p ≤0.05, **p <0.01 and ***p <0.001. Cirrh, cirrhosis; Ctrl, healthy controls; SAH, severe alcoholic hepatitis; sST2, soluble ST2.

and 65.3%, p = 0.12, respectively. Twenty out of the 28 deaths (71.4%) in the SAH group and 6 out of the 8 deaths (75%) in the cirrhosis group were related to infection. In patients with SAH, 2-month survival was higher in those with low concentrations of sST2 (*i.e.* those with serum sST2 lower than 285,355 pg/ml) according to the percentile distribution of sST2 at D0 (\leq 33th percentile, 33–66th percentile and \geq 66th percentile) than in those with intermediate levels (between 285,355 and 625,756 pg/ml) or high levels of sST2 (above 625,756 pg/ml): 89.5% (79.8–99.2) *vs.* 79.5% (66.8–92.2) *vs.* 60% (44.8–64.2), respectively, p = 0.007 (Fig. 1E).

In order to get more insight on the prognostic performance of sST2 in predicting outcome of patients with SAH, we estimated the effect size of the association between sST2 and outcome by calculating HR per 1 SD increase and Harrell's c discrimination index. We also drew ROC curves. sST2 was significantly associated with mortality (HR 1.87; 95% CI 1.3–2.67; *p* <0.0001). Area



Fig. 2. sST2, endotoxemia and infection risk. (A) Levels of circulating 3-HM in the 3 groups (Ctrl n = 12, Cirrh n = 24, SAH n = 119). Results are expressed in pmol/ml. (B) Percentage of patients with SAH developing infection after initiation of medical treatment according the percentile distribution of sST2 at baseline. (C) ROC curve in binormal representation of the capacity of sST2 to predict 2-month infection-free survival in patients with SAH (n = 119). Comparisons between quantitative variables were performed by the Mann-Whitney *U* test or Kurskall-Wallis, as appropriate. n.s., non significant, **p* ≤0.05, ***p* <0.01 and ****p* <0.001. 3-HM, 3-hydroxymyristate; Cirrh, cirrhosis; Ctrl, healthy controls; SAH, severe alcoholic hepatitis; sST2, soluble ST2.

under the ROC curve to predict mortality at 2 months was good: 0.725 (Fig. 1F), as well as Harrell's c-index: 0.7, p < 0.0001. sST2 also predicted 2-month mortality in the subgroup of patients with cirrhosis: HR 1.85, 95% CI 1.07–3.21, p = 0.03 and area under the ROC curve was at 0.778.

sST2 levels correlate with endotoxemia and predict the risk of infection in SAH and in cirrhosis.

In patients with SAH, infection occurred after a median time of 21 (8–39) days following corticosteroid initiation. As expected, quantification of endotoxemia using HPLC-MS/MS showed that patients with SAH had higher levels of 3-HM than patients with cirrhosis and controls: 184.03 *vs.* 156.5 pmol/ml, p = 0.05, and *vs.* 87.8 pmol/ml, p < 0.0001, respectively (Fig. 2A).

Importantly, the probability of developing infection at 2 months increased along with the level of sST2 according to the percentile distribution of sST2 at D0 in SAH: 7.7% in low level vs. 30% in intermediate level (p = 0.01) vs. 57.7% in high level p < 0.0001. The probability of developing an infection was different in those with intermediate vs. high levels of sST2 (p = 0.01). (Fig. 2B). A good performance of sST2 was also found to predict death or infection at 2 months (composite endpoint): HR 1.89, 95% CI 1.46–2.44, p < 0.0001. The subsequent area under the ROC curve was 0.782 (Fig. 2C) and Harrell's c-index was 0.72, p < 0.0001. Finally, when considering the high probability of death in the group of patients with SAH, we determined the Harrell's c-index to predict infection at 2 months, taking death as a competitive event. Performance of sST2 was confirmed with a c-index at 0.73, p < 0.0001. Interestingly, the prognostic value of sST2 to predict death or infection

(composite endpoint) was confirmed in the subgroup of patients with cirrhosis: HR 1.66, 95% CI 1.02–2.71, p = 0.04 and area under the ROC curve was 0.784.

Correlation between sST2, endotoxemia and liver function

We found a weak but significant correlation between sST2 levels and MELD score on the overall cohort of patients with cirrhosis and SAH: Spearman's r = 0.43, *p* <0.0001. This weak correlation was confirmed in the subgroup of patients with SAH: Spearman's r = 0.397, *p* <0.0001. Endotoxemia assessed by serum 3-HM measurement was also weakly associated with sST2 levels on the overall cohort of patients with cirrhosis and SAH: Spearman's r = 0.335, *p* <0.0001. The weak correlation between sST2 and endotoxemia was confirmed in the subgroup of patients with SAH: Spearman's r = 0.331, *p* = 0.0006.

Downstream effectors of IL-33/ST2 (GRK2 and CXCR2) are affected in SAH and their expression is targeted by IL-33 challenge

The kinase GRK2 is a negative regulator of membranous CXCR2 expression and its activity is decreased by the IL-33/ST2 pathway. Immunocytochemistry staining showed increased expression at baseline of GRK2 in circulating PMNs of patients with SAH compared to patients with cirrhosis (p = 0.001) and in patients with cirrhosis compared to controls (p = 0.04): MIRU at 19,527 vs. 12,415 vs. 8,887. In the presence of IL-33, PMNs showed a marked decrease in GRK2 expression (MIRU) in patients with SAH or with cirrhosis, from 19,527 to 14,500, p = 0.009 and from 12,415 to 9,744, p = 0.06, respectively, but not in controls: from 8,887 to 9,632, p = 0.2 (Fig. 3A and 3B).

We then evaluated the expression of the chemokine receptor CXCR2 on the surface of circulating PMNs. CXCR2 is known to be downregulated by GRK2. Consistent with data on GRK2, CXCR2 expression was decreased in the SAH (mean fluorescence index 33.1) and cirrhosis (49.6) groups compared to control (109.3) groups, *p* <0.0001 (Fig. 3B). The difference between cirrhosis and SAH was also significant (*p* = 0.01). Although there was no effect in controls (109.3 *vs.* 106.3, *p* = 0.87), treatment with IL-33 partially restored CXCR2 expression in the SAH (33.1 *vs.* 50.8, *p* <0.001) and cirrhosis (49.6 *vs.* 66.5, *p* = 0.007) groups (Fig. 3C). These results show that CXCR2 expression, a key receptor for neutrophil migration, is downregulated in advanced stages of ALD, especially in SAH. A challenge with IL-33 can restore its expression by decreasing GRK2 activity.

SIGIRR expression using the FACS analysis was not different between controls, patients with cirrhosis or those with SAH, with mean fluorescent intensities (MFIs) of 0.97 vs. 0.99 vs. 1.07, p = 0.8, respectively. SIGIRR expression was also not modified by IL-33 challenge (Fig. S2).

Of note, we did not observe any significant apoptosis upon IL-33 treatment in any group of patients (controls, patients with cirrhosis or with SAH): 0.6% vs. 0.79%, p = 0.6 in controls (n = 4), 1.01% vs. 1.49%, p = 0.11 in cirrhosis (n = 6) and 0.98% vs. 0.83%, p = 0.5 in SAH (n = 5).

Defective PMN migration in SAH: improvement by IL-33

Phagocytosis capacity

The phagocytic capacity of circulating PMNs was evaluated in the 3 groups of patients by a well-validated test (*i.e.* Phagotest). No difference was observed in phagocytic capacity among the 3 groups (MFIs of 63.7, 67.5, and 70.1, respectively, p = 0.8). IL-33



Fig. 3. Expression of molecular effectors of the IL33/sST2 pathway in circulating PMNs. (A) Expression of molecular drivers of the IL33-sST2 pathway in circulating PMNs. Representative immunocytochemistry staining (×200) of intracytoplasmic GRK2 expression in PMNs from Ctrl, Cirrh and SAH patients, with and without pretreatment by IL-33. A picture showing the cytosolic staining of GRK2 in a patient with SAH is provided at a greater magnitude (×400). (B) Quantification of immunocytochemistry staining expressed in MIRUs in the 3 groups (Ctrl n = 7, Cirrh n = 8, SAH n = 7). (C) CXCR2 expression by FACS. Expression of transmembranous chemokine receptor CXCR2 on circulating PMNs of Ctrl (n = 10), Cirrh (n = 15) and SAH (n = 22) patients with or without pretreatment by IL-33. Results are expressed as median MFIs with IQR. Comparisons between quantitative variables were performed by the Mann-Whitney U test, paired-sample Student's t test or Kruskall-Wallis, as appropriate. n.s., non significant, $p \leq 0.05$, p < 0.01 and ***p <0.001. Between 0.05 and 0.1 the p value is indicated. Cirrh, cirrhosis; Ctrl, healthy controls; MFI, mean fluorescent intensity; MIRU, mean intensity with relative units; PMNs, polymorphonuclear neutrophils; SAH, severe alcoholic hepatitis; sST2, soluble ST2. (This figure appears in color on the web.)

pretreatment did not significantly change the phagocytic capacity in the 3 groups (Fig. 4A).

Migration capacity

The migration capacity of PMNs driven by IL-8 was assessed using transwell assay experiments, in the presence or absence of IL-33. Spontaneous migration (in the absence of cytokines) was low at baseline and no difference was observed between the SAH,



Fig. 4. Evaluation of migration function and phagocytosis of circulating PMNs during ALD. (A) Migration capacity of circulating PMNs of Ctrl (n = 10), Cirrh (n = 15) and SAH (n = 22) patients has been evaluated by transwell assays in the presence or absence of IL-33. Results are expressed by median percentage (and IQR) of migrated cells after standard deposition in the upper well of 1×10^6 cells. (B) Phagocytic capacity was evaluated by standard test with FACS. Results are expressed as median in MFIs with IQR in Ctrl (n = 3), Cirrh (n = 10) and SAH (n = 3) patients. Comparisons between quantitative variables were performed by the Mann-Whitney *U* test, paired-sample Student's *t* test or Kruskall-Wallis, as appropriate. n.s., non significant, * $p \le 0.05$, **p < 0.01 and ***p < 0.001. Cirrh, cirrhosis; Ctrl, healthy controls; MFI, mean fluorescent intensity; PMNs, polymorphonuclear neutrophils; SAH, severe alcoholic hepatitis; sST2, soluble ST2.

cirrhosis and control groups: 0.14%, 0.15%, 0.17%, respectively (p = 0.52) (Fig. 4B). Treatment with IL-8 increased the percentage of migrating PMNs across the 3 groups, but the effect of IL-8 was less marked in the SAH group (3.7%, p = 0.0001) than in cirrhosis (7.3%, p = 0.0003) or control (12.2%, p = 0.04) groups (Fig. 4B).

Treatment with IL-33 had no effect on the percentage of migrating cells in controls (12.2% vs. 13.1%, p = 0.7). In contrast, treatment with IL-33 increased the percentage of migrating PMNs compared to IL-8 treatment alone in the SAH and cirrhosis groups, showing partial restoration of the migration defect: SAH (3.7% vs. 4.7%, p = 0.0009) and cirrhosis (7.3% vs. 8.7%, p = 0.006) (Fig. 4B).

We finally analyzed the impact of response to treatment on the changes in PMN migration between the initiation of prednisolone (D0) and day 7 of therapy (D7) in SAH. IL-8 increased the percentage of migrating PMNs on D0 in responders (*i.e.* Lille <0.45): 0.19% vs. 4.02%, p = 0.002. This effect was further improved by IL-33: 4.02% vs. 8.33%, p = 0.006. Similarly, in nonresponders, IL-8 increased the percentage of migrating PMNs on D0 compared to baseline (0.13% vs. 2.08%, p = 0.03) and the addition of IL-33 further increased this percentage (2.08% vs.

NAFLD and Alcohol-Related Liver Diseases



Α

В

Fig. 5. Evaluation of migration capacity of PMNs in SAH according to response to treatment. Migration capacity of circulating PMNs of patients with SAH (n = 22) has been performed by transwell assays in the presence or absence of IL-33. Results are expressed as median percentage (and IQR) of migrated cells after standard deposition in the upper well of 1×10^6 cells. (A) Evolution of migration capacity of circulating PMNs at D0 according to response to treatment (responders n = 11, non-responders n = 11). (B) Evolution of migration capacity of circulating PMNs at D7 according to response to treatment. Comparisons between quantitative variables were performed by the Mann-Whitney *U* test, paired-sample Student's *t* test or Kruskall-Wallis, as appropriate. n.s., non significant, * $p \le 0.05$, **p < 0.01 and ***p < 0.001. Between 0.05 and 0.1 the p value is indicated. PMNs, polymorphonuclear neutrophilis; SAH, severe alcoholic hepatitis.

3.31%, p = 0.07). The percentage of migrated cells at D0 following treatment with IL-8 was similar in responders and non-responders (p = 0.3) but IL-33 induced greater migration of PMNs in responders than in non-responders (p = 0.05) (Fig. 5A).

This difference in migration between responders and nonresponders was greater after 7 days of treatment with steroids. Indeed, at D7, treatment with IL-8 increased the percentage of migrating PMNs (0.11% vs. 4.93%, p = 0.0003) in responders and this percentage increased again with the addition of IL-33 (4.93% vs. 11.22% p = 0.0005). At D7, IL-8 and IL-33 also increased PMN migration in non-responders: (addition of IL-8: 0.08% vs. 1.51%, p = 0.03; addition of IL-33: 1.51% vs. 3.8%, p = 0.005). After 7 days of steroids, the difference became significant between responders and non-responders after challenge with IL-8 (4.93% vs. 1.51%, p = 0.04). Furthermore, treatment with IL-33 was more effective in PMNs from responders than in those from nonresponders (11.22% vs. 3.8%, p = 0.005) (Fig. 5B). In order to

assess if larger amounts of IL-33 could overcome the less good response to IL-33 in non-responders, we performed migration assays with increased doses of IL-33 (50, 200, 500 and 1,000 ng/ ml) in both responders and non-responders at baseline and after 7 days of prednisolone course. There was no improvement in migration capacity in both groups with the increased doses compared to the standard dose of 50 ng/ml (data not shown).

Discussion

Alcoholic hepatitis is associated with a high incidence of infection, a major driver of short-term mortality. While several pathways have been investigated in decompensated cirrhosis⁹ there are fewer data in alcoholic hepatitis. This study shows that the IL-33/ST2 pathway is altered in PMNs from patients with SAH and that this defect can be restored, at least in part, by IL-33. Indeed, the levels of circulating sST2, a biomarker indicative of IL-33/ST2 pathway activity, are associated with short-term mortality, development of infection and response to steroids. We also observed that different downstream effectors of the IL33/ST2 pathway were altered in SAH. Indeed, SAH is associated with an increased expression of GRK2 and decreased expression of CXCR2, which can be restored by challenging PMNs with IL-33. Decreased CXCR2 expression can be linked to a reduced migration capacity induced by IL-8, which can be improved by challenge with IL-33. The main study findings are shown in Fig. 6.

The issue of controls is critical in translational research for alcoholic hepatitis. Indeed, most patients admitted with SAH (*i.e.* with a Maddrey discriminant function >32) have underlying cirrhosis. Thus, we used 2 different control groups (*i.e.* patients with cirrhosis and healthy controls) to prove that changes in the IL-33/ST2 pathway in SAH are related to alcoholic hepatitis rather than decompensated cirrhosis. Comparison between cirrhosis and controls also shows the defect in this pathway during liver decompensation without any inflammatory process.

Endotoxemia was quantified using the HPLC-MS/MS method to determine serum concentrations of 3-HM. We chose this technique because HPLC-MS/MS can more accurately identify patients with low-grade endotoxemia, compared to the classical LAL (limulus amebocyte lysate) test, due to its high specificity and low limits of detection and quantification. Moreover, the measurement is not affected by the lipid content of samples, thus avoiding bias due to lipid-rich particles observed with the LAL test.²³ Such a technique has been validated in patients with cirrhosis.²²

Herein, we confirm that patients with SAH have higher endotoxemia than patients with cirrhosis and controls.³⁰ The weak although significant correlation between sST2 levels and MELD score or endotoxemia suggests that liver failure and latent infection do not play an exclusive role to drive sST2 up in patients with SAH. Compared to the study by Alves-Filho *et al.*²⁰ we think that it is not only infection or liver insufficiency that inhibits the IL33-ST2 pathway but rather that this defect is a characteristic of SAH. Future studies are required to determine if systemic inflammation associated with SAH plays a role in the defect in the IL-33/ST2 pathway.

SIGIRR has been suggested to be a downregulator of the Tolllike receptor pathways implicated in sepsis in murine models, especially in neutrophils, with conflicting results in terms of functional consequences.^{31,32} In the present study, we did not observe any significant difference in SIGIRR expression between



Fig. 6. Main study findings. (This figure appears in color on the web.)

controls, patients with cirrhosis and those with SAH, in the presence or the absence of IL-33.

Because the incidence of infection is a major driver of mortality in SAH,^{5,8,33} there is an urgent need for reliable markers able to identify patients at a greater risk of becoming infected. Although baseline severity scores have been combined with scores evaluating medical management response, the percentage of well-classified patients is still around 80%. Thus, the outcome of some patients is not properly predicted in close to 20% of cases. Moreover, in patients initially classified as having good prognosis and who respond to therapy, the occurrence of an infection is responsible for a major drop in survival.⁸ Circulating sST2 has been proposed as a biomarker in several diseases^{34–3} and adding this marker to available scoring systems could optimize the prediction of clinical events in SAH. We show here using a large sample size and a robust methodology that sST2 not only predicts mortality but also the probability of getting infected in SAH and in cirrhosis, even though circulating levels of sST2 are different between the 2 conditions. Such results are promising because there is a large unmet need for biomarkers to predict the development of infection in both SAH and cirrhosis. Since elevated baseline sST2 values are associated with the risk of development of infection after treatment, patients with high sST2 might be candidates for trials testing strategies against infection. We observed higher levels of sST2 after 7 days of prednisolone exposure in both responders and non-responders to corticosteroids. It has already been reported that sST2 levels are increased by steroid treatment in ulcerative colitis and in systemic lupus erythematosus^{37–39} and we feel that its evolution under prednisolone is not indicative of the global activity of the IL-33/ST2 pathway.

There has been no significant improvement in the pharmaceutical management of SAH in the past few years. Based on the knowledge that infection is associated with a high short-term probability of death, results in trials testing antibiotic prophylaxis are expected (Rifaximin: NCT02116556; Ciprofloxacin: NCT02326103; Amoxicillin and Clavulanate: NCT02281929). While strategies with antibiotics are promising, they do not restore the immune defect associated with SAH. Stimulating

Research Article

PMN migration to the site of infection with IL-33 might represent an appealing option. Indeed, alterations in migration capacity have been associated with an increased risk of infection in humans.⁴⁰ Interestingly an increased migration capacity of circulating PMNs with IL-33 was found in both responders and non-responders to treatment in patients with SAH. We previously showed that non-responders to steroids are more likely to develop infection than responders.⁸ The present study shows that PMNs from non-responders have a lower migration capacity following a challenge with IL-8 than those from responders and that IL-33 leads to an improvement in migration capacity in both groups. Although PMNs from non-responders are less responsive to IL-33, this cytokine improves migration capacity. This is particularly important because of the high rate of infection in these patients. Furthermore, PMNs from non-responders are less responsive to IL-33 at D0, suggesting that the defect in migration in these patients is already present in part before steroid exposure. The defect in the IL-33/ST2 pathway in the neutrophils of non-responders to corticosteroids seems to be related to higher levels of the decoy receptor sST2 but also to a less good response to IL-33 than responders that cannot be improved with higher doses of IL-33.

While neutrophils were treated with a dose of 50 ng/ml in the *ex vivo* experiments, the detected dose in the serum was around 100 pg/ml. Such a difference can be viewed as a supraphysiological stimulus. Other authors have also used similar doses.^{20,41–44} In addition, we cannot exclude that the detection of IL-33 in the serum of patients lacks sensitivity. Based on our results, we propose that in patients with SAH, restoration of the IL-33/ST2 pathway may help prevent and treat infection in both responders and non-responders.

In the present study, the IL-33/ST2 pathway did not modulate the phagocytic capacity of circulating PMNs suggesting that this pathway only improves PMNs influx at the site of infection. We would like to underline that impairment in neutrophil function during ALD is not restricted to decreased phagocytosis but also includes defects in microbial killing with several pathways involved such as myeloperoxidase, TIM-3, PD-1, *etc.*⁹ Thus, future studies targeting pathways involved in phagocytic capacity in circulating PMNs are needed.

Alcoholic hepatitis is a complex disease and the role of PMNs is not fully understood. Indeed, there is a paradox between the liver infiltration by neutrophils (suggesting that activation of these cells is harmful to the liver) and a defect in neutrophil migration capacity (suggesting that a defect in PMN function is harmful to infection). In addition, IL-8 is the most upregulated cytokine in patients with acute drinking/alcoholic hepatitis vs. patients without SAH in the CANONIC study evaluating the profile of patients with acute-on-chronic liver failure.⁴⁵ IL-8 is also elevated in the blood and in the liver of patients with SAH in the study by Taïeb et al.⁴⁶ Based on our results on neutrophil migration upon IL-8 and IL-33, we cannot exclude that promoting the IL-33/ST2 pathway in patients with ALD could result in a higher recruitment of neutrophils in the liver. Although the present study does not address this issue, it can be hypothesized that neutrophil migration triggers are different for the liver and the site of infection. The discrepancies between migration in the liver and to the site of infection could be related to different patterns of chemokine expression. However, IL-8 is not the only signal leading to neutrophil recruitment, as reviewed by Gustot et al.⁹ Specific studies testing chemotaxis targets are needed for alcoholic hepatitis.

In conclusion, this study shows that circulating neutrophil migration is altered in SAH and that this defect is reversible by restoring the IL-33/ST2 pathway. Circulating sST2 levels are associated with survival and the probability of infection. Our results suggest that treatment with IL-33 represents a new potential therapeutic strategy to decrease the risk of infection in SAH.

Abbreviations

3-HM, 3-hydroxymyristate; ALD, alcohol-related liver disease; AH, alcoholic hepatitis; Cirrh, cirrhosis; Ctrls, controls; DAMP, damage-associated molecular pattern; HR, hazard ratio; IFN, interferon; LPS, lipopolysaccharide; MELD, model for end-stage liver disease; MFI, mean fluorescent intensity; MIRU, mean intensity with relative units; PMN, polymorphonuclear neutrophils; SAH, severe alcoholic hepatitis; sST2, soluble ST2.

Financial support

The authors would like to thank the Association Française pour l'Etude du Foie (AFEF, French society of hepatology, France), the National Institute on Alcohol Abuse and Alcoholism (NIAAA, United States, grant ID 1U01AA021908) and FAPEMIG Région Hauts-de-France, France for the financial support.

Conflict of interest

The authors declare no conflicts of interest that pertain to this work.

Please refer to the accompanying ICMJE disclosure forms for further details.

Authors' contributions

Design of the study: FA, PM, LD, AL. Acquisition of data: FA, MBS, FM, GL, MN, JD, LCNW, JPPDB, DGH, SD, EG, AP, SCM, PM, LD, AL. Statistical analysis: FA, JL, ED, AL. Drafting of the manuscript and critical review: FA, RB, PM, LD, AL.

Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jhep.2019.12.017.

References

- Maddrey WC, Boitnott JK, Bedine MS, Weber FL, Mezey E, White RI. Corticosteroid therapy of alcoholic hepatitis. Gastroenterology 1978;75:193–199.
- [2] Mathurin P, Louvet A, Duhamel A, Nahon P, Carbonell N, Boursier J, et al. Prednisolone with vs without pentoxifylline and survival of patients with severe alcoholic hepatitis: a randomized clinical trial. JAMA 2013;310:1033–1041.
- [3] Lucey MR, Schaubel DE, Guidinger MK, Tome S, Merion RM. Effect of alcoholic liver disease and hepatitis C infection on waiting list and posttransplant mortality and transplant survival benefit. Hepatology 2009;50:400–406.
- [4] Mathurin P, O'Grady J, Carithers RL, Phillips M, Louvet A, Mendenhall CL, et al. Corticosteroids improve short-term survival in patients with severe alcoholic hepatitis: meta-analysis of individual patient data. Gut 2011;60:255–260.
- [5] Thursz MR, Richardson P, Allison M, Austin A, Bowers M, Day CP, et al. Prednisolone or pentoxifylline for alcoholic hepatitis. N Engl J Med 2015;372:1619–1628.
- [6] Louvet A, Thursz MR, Kim DJ, Labreuche J, Atkinson SR, Sidhu SS, et al. Corticosteroids reduce risk of death within 28 days for patients with severe alcoholic hepatitis, compared with pentoxifylline or placebo-a metaanalysis of individual data from controlled trials. Gastroenterology 2018;155:458–468.e8.
- [7] Louvet A, Naveau S, Abdelnour M, Ramond M-J, Diaz E, Fartoux L, et al. The Lille model: a new tool for therapeutic strategy in patients with



severe alcoholic hepatitis treated with steroids. Hepatology 2007;45:1348–1354.

- [8] Louvet A, Wartel F, Castel H, Dharancy S, Hollebecque A, Canva-Delcambre V, et al. Infection in patients with severe alcoholic hepatitis treated with steroids: early response to therapy is the key factor. Gastroenterology 2009;137:541–548.
- [9] Gustot T, Fernandez J, Szabo G, Albillos A, Louvet A, Jalan R, et al. Sepsis in alcohol-related liver disease. J Hepatol 2017;67:1031–1050.
- [10] Mookerjee RP, Stadlbauer V, Lidder S, Wright GAK, Hodges SJ, Davies NA, et al. Neutrophil dysfunction in alcoholic hepatitis superimposed on cirrhosis is reversible and predicts the outcome. Hepatology 2007;46:831–840.
- [11] Stadlbauer V, Mookerjee RP, Wright GA, Davies NA, Jürgens G, Hallström S, et al. Role of Toll-like receptors 2, 4, and 9 in mediating neutrophil dysfunction in alcoholic hepatitis. Am J Physiol Gastrointest Liver Physiol 2009;296:G15–G22.
- [12] Boussif A, Rolas L, Weiss E, Bouriche H, Moreau R, Périanin A. Impaired intracellular signaling, myeloperoxidase release and bactericidal activity of neutrophils from patients with alcoholic cirrhosis. J Hepatol 2016;64:1041–1048.
- [13] Fiuza C, Salcedo M, Clemente G, Tellado JM. In vivo neutrophil dysfunction in cirrhotic patients with advanced liver disease. J Infect Dis 2000;182:526–533.
- [14] Moussion C, Ortega N, Girard J-P. The IL-1-like cytokine IL-33 is constitutively expressed in the nucleus of endothelial cells and epithelial cells in vivo: a novel "alarmin"? PLoS One 2008;3:e3331.
- [15] Hudson CA, Christophi GP, Gruber RC, Wilmore JR, Lawrence DA, Massa PT. Induction of IL-33 expression and activity in central nervous system glia. J Leukoc Biol 2008;84:631–643.
- [16] Préfontaine D, Lajoie-Kadoch S, Foley S, Audusseau S, Olivenstein R, Halayko AJ, et al. Increased expression of IL-33 in severe asthma: evidence of expression by airway smooth muscle cells. J Immunol 2009;183:5094– 5103.
- [17] Arshad MI, Piquet-Pellorce C, L'Helgoualc'h A, Rauch M, Patrat-Delon S, Ezan F, et al. TRAIL but not FasL and TNFα, regulates IL-33 expression in murine hepatocytes during acute hepatitis. Hepatology 2012;56:2353– 2362.
- [18] Wang M, Shen G, Xu L, Liu X, Brown JM, Feng D, et al. IL-1 receptor like 1 protects against alcoholic liver injury by limiting NF-κB activation in hepatic macrophages. J Hepatol 2017;68:109–117.
- [19] Sun Z, Chang B, Gao M, Zhang J, Zou Z. IL-33-ST2 axis in liver disease: progression and challenge. Mediators Inflamm 2017;2017:5314213.
- [20] Alves-Filho JC, Sônego F, Souto FO, Freitas A, Verri WA, Auxiliadora-Martins M, et al. Interleukin-33 attenuates sepsis by enhancing neutrophil influx to the site of infection. Nat Med 2010;16:708–712.
- [21] Lucey MR, Mathurin P, Morgan TR. Alcoholic hepatitis. N Engl J Med 2009;360:2758–2769.
- [22] Weil D, Pais de Barros J-P, Mourey G, Laheurte C, Cypriani B, Badet N, et al. Circulating levels of 3-hydroxymyristate, a direct quantification of endotoxaemia in noninfected cirrhotic patients. Liver Int 2019;39:106– 114.
- [23] Pais de Barros J-P, Gautier T, Sali W, Adrie C, Choubley H, Charron E, et al. Quantitative lipopolysaccharide analysis using HPLC/MS/MS and its combination with the limulus amebocyte lysate assay. J Lipid Res 2015;56:1363–1369.
- [24] Jensen EC. Quantitative analysis of histological staining and fluorescence using ImageJ. Anat Rec (Hoboken) 2013;296:378–381.
- [25] Sebastiani S, Allavena P, Albanesi C, Nasorri F, Bianchi G, Traidl C, et al. Chemokine receptor expression and function in CD4+ T lymphocytes with regulatory activity. J Immunol 2001;166:996–1002.
- [26] Traidl-Hoffmann C, Kasche A, Jakob T, Huger M, Plötz S, Feussner I, et al. Lipid mediators from pollen act as chemoattractants and activators of polymorphonuclear granulocytes. J Allergy Clin Immunol 2002;109:831– 838.
- [27] Pionnier N, Brotin E, Karadjian G, Hemon P, Gaudin-Nomé F, Vallarino-Lhermitte N, et al. Neutropenic mice provide insight into the role of skin-

infiltrating neutrophils in the host protective immunity against filarial infective larvae. PLoS Negl Trop Dis 2016;10:e0004605.

- [28] Portney L, Watkins M. Foundations of Clinical Research: Applications to Practice. 2nd Edition. Upper Saddle River: Prentice Hall; 2000.
- [29] Wolbers M, Koller M, Witteman J, Steyerberg E. Prognostic models with competing risks: methods and application to coronary risk prediction. Epidemiology 2009;20(4):555–561.
- [30] Michelena J, Altamirano J, Abraldes JG, Affò S, Morales-Ibanez O, Sancho-Bru P, et al. Systemic inflammatory response and serum lipopolysaccharide levels predict multiple organ failure and death in alcoholic hepatitis. Hepatology 2015;62:762–772.
- [31] Ueno-Shuto K, Kato K, Tasaki Y, Sato M, Sato K, Uchida Y, et al. Lipopolysaccharide decreases single immunoglobulin interleukin-1 receptorrelated molecule (SIGIRR) expression by suppressing specificity protein 1 (Sp1) via the Toll-like receptor 4 (TLR4)-p38 pathway in monocytes and neutrophils. J Biol Chem 2014;289:18097–18109.
- [32] Blok DC, van Lieshout MH, Hoogendijk AJ, Florquin S, de Boer OJ, Garlanda C, et al. Single immunoglobulin interleukin-1 receptor-related molecule impairs host defense during pneumonia and sepsis caused by Streptococcus pneumoniae. J Innate Immun 2014;6:542–552.
- [33] Vergis N, Atkinson SR, Knapp S, Maurice J, Allison M, Austin A, et al. In patients with severe alcoholic hepatitis, prednisolone increases susceptibility to infection and infection-related mortality, and is associated with high circulating levels of bacterial DNA. Gastroenterology 2017;152:1068–1077.e4.
- [34] Coronado MJ, Bruno KA, Blauwet LA, Tschöpe C, Cunningham MW, Pankuweit S, et al. Elevated sera sST 2 is associated with heart failure in men ≤50 years old with myocarditis. J Am Heart Assoc 2019;8:e008968.
- [35] Alam ML, Katz R, Bellovich KA, Bhat ZY, Brosius FC, de Boer IH, et al. Soluble ST2 and Galectin-3 and progression of CKD. Kidney Int Rep 2019;4:103–111.
- [36] Watanabe M, Nakamoto K, Inui T, Sada M, Honda K, Tamura M, et al. Serum sST2 levels predict severe exacerbation of asthma. Respir Res 2018;19:169.
- [37] Díaz-Jiménez D, Núñez L, De la Fuente M, Dubois-Camacho K, Sepúlveda H, Montecino M, et al. A functional IL1RL1 variant regulates corticosteroid-induced sST2 expression in ulcerative colitis. Sci Rep 2017;7:10180.
- [38] Díaz-Jiménez D, Núñez LE, Beltrán CJ, Candia E, Suazo C, Alvarez-Lobos M, et al. Soluble ST2: a new and promising activity marker in ulcerative colitis. World J Gastroenterol 2011;17:2181–2190.
- [39] Mok MY, Huang FP, Ip WK, Lo Y, Wong FY, Chan EYT, et al. Serum levels of IL-33 and soluble ST2 and their association with disease activity in systemic lupus erythematosus. Rheumatolology (Oxford) 2010;49:520–527.
- [40] Egger G, Aigner R, Glasner A, Hofer HP, Mitterhammer H, Zelzer S. Blood polymorphonuclear leukocyte migration as a predictive marker for infections in severe trauma: comparison with various inflammation parameters. Intensive Care Med 2004;30:331–334.
- [41] Bandara G, Beaven MA, Olivera A, Gilfillan AM, Metcalfe DD. Activated mast cells synthesize and release soluble ST2-a decoy receptor for IL-33. Eur J Immunol 2015;45:3034–3044.
- [42] Hayakawa H, Hayakawa M, Kume A, Tominaga S. Soluble ST2 blocks interleukin-33 signaling in allergic airway inflammation. J Biol Chem 2007;282:26369–26380.
- [43] Le HT, Tran VG, Kim W, Kim J, Cho HR, Kwon B. IL-33 priming regulates multiple steps of the neutrophil-mediated anti-Candida albicans response by modulating TLR and dectin-1 signals. J Immunol 2012;189:287–295.
- [44] Lan F, Yuan B, Liu T, Luo X, Huang P, Liu Y, et al. Interleukin-33 facilitates neutrophil recruitment and bacterial clearance in S. aureus-caused peritonitis. Mol Immunol 2016;72:74–80.
- [45] Clària J, Stauber RE, Coenraad MJ, Moreau R, Jalan R, Pavesi M, et al. Systemic inflammation in decompensated cirrhosis: characterization and role in acute-on-chronic liver failure. Hepatology 2016;64:1249–1264.
- [46] Taïeb J, Mathurin P, Elbim C, Cluzel P, Arce-Vicioso M, Bernard B, et al. Blood neutrophil functions and cytokine release in severe alcoholic hepatitis: effect of corticosteroids. J Hepatol 2000;32:579–586.